Author’s response to reviews

Title: Spectrum of PAH gene variants among a population of Han Chinese patients with phenylketonuria from Northern China

Authors:
Ning Liu (liuning903@163.com)
Qiuying Huang (hqying@xmu.edu.cn)
Qingge Li (liqingge56@126.com)
Dehua Zhao (zhaodh63@163.com)
Xiaole Li (lixiaole8711@163.com)
Lixia Cui (lixiacui60@163.com)
Ying Bai (baiying198611@126.com)
Yin Feng (wuqh7703@163.com)
Xiangdong Kong (kongxd1968@163.com)

Version: 5 Date: 07 Sep 2017

Author’s response to reviews:

1. We note that the Authors' Contributions still have not changed for HQ and LQ. As stated previously, according to the ICMJE guidelines, to qualify as an author one should have helped to write or revise the manuscript, as well as provide or analyse data. Please clarify whether these two authors helped to write or revise the manuscript at all and include this information in the Authors' Contributions section. Please see the ICMJE guidelines for all the requirements for authorship:  http://www.icmje.org/recommendations/browse/roles-and-responsibilities/defining-the-role-of-authors-and-contributors.html#two

Response: Thank you. We have read the the ICMJE guidelines for all the requirements for authorship. Per your request, we we have clarified the Authors' contributions section in the revised article on page 14, line 268-269.

2. Please include a Figure Legend heading above Figure 1.
Response: Thank you for your comment. We have included a Figure Legend heading above Figure 1.